Author's response to reviews

Title: TrkA is amplified in malignant melanoma patients and induces an anti-proliferative response in cell lines

Authors:

Luigi Pasini (luigi.pasini@unitn.it)
Angela Re (angela.re@unitn.it)
Toma Tebaldi (t.tebaldi@unitn.it)
Gianluca Ricci (gianluca.ricci@unitn.it)
Sebastiana Boi (sebastiana.boi@apss.tn.it)
Valentina Adami (valentina.adami@unitn.it)
Mattia Barbareschi (mattia.barbareschi@apss.tn.it)
Alessandro Quattrone (alessandro.quattrone@unitn.it)

Version: 9
Date: 1 September 2015

Author's response to reviews: see over
Dear Dr. Natan,

Thank you for having accepted the manuscript for publication in BMC Cancer and for your continued attention. We have revised the final version of the paper by introducing the emendations requested. Specifically, we have removed patients’ gender information and changed the age to “age range”, in Table S1. We have also included the statement “All authors have read and approved the manuscript”, at the end of Authors' contributions section.

Yours sincerely,

Luigi Pasini and Alessandro Quattrone